Author’s response to reviews

Title: Prevalence and correlates of positive mental health in Chinese adolescents

Authors:

Cheng Guo (cheng.guo@ki.se)
Göran Tomson (goran.tomson@ki.se)
Christina Keller (christina.keller@ju.se)
Fredrik Söderqvist (fredrik.soderqvist@ltv.se)

Version: 4 Date: 22 Jan 2018

Author’s response to reviews:

The revised manuscript and the ethical approval for this study in both Chinese and English are attached.